Author’s response to reviews

Title: Congenital partial arhinia: a case report

Authors:

Guzin Akkuzu (guzina@baskent-ank.edu.tr)
Babur Akkuzu (babura@baskent-ank.edu.tr)
Erdinc Aydin (erdinca@baskent-ank.edu.tr)
Murat Derbent (muratd@baskent-ank.edu.tr)
Levent Ozluoglu (levento@baskent-ank.edu.tr)

Version: 4 Date: 24 July 2007

Author's response to reviews:

The title has been changed as Congenital Partial Arhinia as advised by referee 2.

Last sentence of discussion is changed as "We considered our patient as having partial arhinia because there were some remnants of the external nose and a cul-de-sac accompanied by central nervous system and ear anomalies, a submucosal cleft palate, and hypotelorism."

A MRI investigation has not been performed at that time and is not possible any more. So our central nervous system findings are limited to CT sections. Unfortunately the presented sections are the best available ones.

The article is read and corrected by a native English speaking editor, so there are multiple grammar and spelling corrections all through out the manuscript.